

Rift Valley fever in northern Senegal: A modelling approach to analyse the processes underlying virus circulation recurrence

Benoit Durand, Moustapha Lo Modou, Annelise Tran, Aminata Ba, Fafa Sow, Jaber Belkhiria, Assane Gueye Fall, Biram Biteye, Vladimir Grosbois,

Véronique Chevalier

► To cite this version:

Benoit Durand, Moustapha Lo Modou, Annelise Tran, Aminata Ba, Fafa Sow, et al.. Rift Valley fever in northern Senegal: A modelling approach to analyse the processes underlying virus circulation recurrence. PLoS Neglected Tropical Diseases, 2020, 14 (6), pp.e0008009. 10.1371/journal.pntd.0008009 . hal-02934597

HAL Id: hal-02934597 https://hal.inrae.fr/hal-02934597v1

Submitted on 9 Sep 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



GOPEN ACCESS

Citation: Durand B, Lo Modou M, Tran A, Ba A, Sow F, Belkhiria J, et al. (2020) Rift Valley fever in northern Senegal: A modelling approach to analyse the processes underlying virus circulation recurrence. PLoS Negl Trop Dis 14(6): e0008009. https://doi.org/10.1371/journal.pntd.0008009

Editor: Dennis A. Bente, University of Texas Medical Branch, UNITED STATES

Received: December 18, 2019

Accepted: April 22, 2020

Published: June 1, 2020

Copyright: © 2020 Durand et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This study was supported by Vmerge project (Emerging viral vector-borne diseases) and by the IIe-de-France Region as part of the DIM-1Health project. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Rift Valley fever in northern Senegal: A modelling approach to analyse the processes underlying virus circulation recurrence

Benoit Durand¹*, Moustapha Lo Modou², Annelise Tran^{3,4,5}, Aminata Ba², Fafa Sow², Jaber Belkhiria⁶, Assane Gueye Fall², Biram Biteye², Vladimir Grosbois³, Véronique Chevalier^{4,7,8}*

 Epidemiology Unit, Laboratory for Animal Health, French Agency for Food, Environmental and Occupational Health and Safety (ANSES), University Paris-Est, Maisons-Alfort, France, 2 Institut Sénégalais de Recherche Agricole (ISRA), Dakar, Sénégal, 3 CIRAD, UMR TETIS, Sainte-Clotilde, Réunion, France, 4 CIRAD, UMR ASTRE, Montpellier, France, 5 Université Montpellier, Montpellier, France, 6 Center for Animal Disease Modeling and Surveillance, Department of Medicine & Epidemiology, School of Veterinary Medicine, University of California Davis, California, United States of America, 7 CIRAD, UMR ASTRE, Phnom Penh, Cambodia, 8 Epidemiology and Public Health Unit, Institut Pasteur du Cambodge, Phnom Penh, Cambodia

* benoit.durand@anses.fr (BD); veronique.chevalier@cirad.fr (VC)

Abstract

Rift Valley fever (RVF) is endemic in northern Senegal, a Sahelian area characterized by a temporary pond network that drive both RVF mosquito population dynamics and nomadic herd movements. To investigate the mechanisms that explain RVF recurrent circulation, we modelled a realistic epidemiological system at the pond level integrating vector population dynamics, resident and nomadic ruminant herd population dynamics, and nomadic herd movements recorded in Younoufere area. To calibrate the model, serological surveys were performed in 2015–2016 on both resident and nomadic domestic herds in the same area. Mosquito population dynamics were obtained from a published model trained in the same region. Model comparison techniques were used to compare five different scenarios of virus introduction by nomadic herds associated or not with vertical transmission in Aedes vexans. Our serological results confirmed a long lasting RVF endemicity in resident herds (IgG seroprevalence rate of 15.3%, n = 222), and provided the first estimation of RVF IgG seroprevalence in nomadic herds in West Africa (12.4%, n = 660). Multivariate analysis of serological data suggested an amplification of the transmission cycle during the rainy season with a peak of circulation at the end of that season. The best scenario of virus introduction combined yearly introductions of RVFV from 2008 to 2015 (the study period) by nomadic herds, with a proportion of viraemic individuals predicted to be larger in animals arriving during the 2nd half of the rainy season (3.4%). This result is coherent with the IgM prevalence rate (4%) found in nomadic herds sampled during the 2nd half of the rainy season. Although the existence of a vertical transmission mechanism in Aedes cannot be ruled out, our model demonstrates that nomadic movements are sufficient to account for this endemic circulation in northern Senegal.

Author summary

Rift Valley fever (RVF) is one of the most important vector borne disease in Africa, seriously affecting the health of domestic ruminants and humans and leading to severe economic consequences. This disease is endemic in northern Senegal, a Sahelian area characterized by a temporary pond network that drive both RVF mosquito population dynamics and nomadic herd movements. Two non-exclusive mechanisms may support this endemicity: recurrent introductions of the virus by nomadic animals, and vertical transmission of the virus (i.e. from infected female mosquito to eggs) in local Aedes populations. The authors followed resident and nomadic domestic herds for 1 year. They used the data thus obtained to model a realistic epidemiological system at the pond level integrating vector population dynamics, resident and nomadic ruminant herd population dynamics. They found that the best scenario explaining RVF remanence combined yearly introductions of RVFV by nomadic herds, with a viraemic proportion predicted to be larger in animals arriving during the 2nd half of the rainy season, which is consistent with an amplification of virus circulation in the area during the rainy season. Although the existence of a vertical transmission mechanism in Aedes cannot be ruled out, their results demonstrates that nomadic movements are sufficient to account for this endemic circulation in northern Senegal.

Introduction

Vector borne pathogen circulation results from complex interactions between hosts, vectors and pathogens. These interactions are modulated by intrinsic factors such as the genetic characteristics of vectors and hosts, but also by factors influencing larval habitat density and host numbers, such as irrigation, rainfall, human population density, animal transhumance and trade, or social practices [1, 2]. Understanding this complexity is essential for controlling vector borne diseases and mitigating their impacts.

Rift Valley fever (RVF), is one of the most important vector borne disease in Africa [3], seriously affecting the health of domestic ruminants and humans and leading to severe economic consequences [4, 5]. RVF is an acute, viral disease, caused by a Phlebovirus (Bunyaviridae family) [3]. Infection leads to abortions in pregnant animals and high mortalities in new born sheep and goats. Humans get infected after being in contact with infectious tissues -blood, body fluids or abortion products, or through the bite of an infectious mosquito. Most human cases are moderate, but rare severe complications such as hepatitis or encephalitis could occur and lead to death. Transmission between ruminants occurs via mosquito bites, mainly from mosquitoes in the genera *Aedes*, *Culex* and *Mansonia* [6, 7], and probably through direct contact with infectious animal tissues and fluids [8, 9].

Since its first recognition in 1931 in Kenya, RVF outbreaks have been regularly reported in sub-Saharan Africa [10–16], the Indian Ocean [17, 18] and the Arabian Peninsula [19]. The most recent outbreaks occurred in Mauritania (2015), Niger (2016), Uganda (2016) and Mayotte (2018–2019) [10, 20–22].

In Senegal, West Africa, RVF is endemic and has been repeatedly reported among humans, livestock, and mosquitoes, especially in the Ferlo (northern Senegal) [23–27]. The Ferlo is a typical sahelian area, with a semi–arid climate, annual rainfall ranging from 300 to 500 mm and a rainy season lasting from July to October/November. This area is sprinkled by a complex and dense network of temporary ponds located within the fossil Ferlo river bed. These ponds

fill up during the rainy season and dry out during the rest of the year. During the rainy season, pond water levels show daily fluctuations, increasing with rainfall and decreasing with infiltration (favoured by sandy-loam soils), high evapotranspiration and water consumption by live-stock and humans. This water availability combined with high quality grass, attract a massive flow of nomadic breeders coming from the whole country with their family and herd, and sometimes from neighbouring countries [28]. Nomadic herds settle around these temporary ponds for a couple of weeks or sometimes several months, depending on grass and water availability [28]. In the middle of dry seasons, ponds are totally dried, and only large wells can be used for ruminants.

Temporary ponds are also important breeding sites for RVFV mosquito vectors, i.e. Culex poicilipes (Theobald) and Aedes vexans (Meigen) [24, 29]. Their respective population dynamics are strongly linked to the pond water surface dynamics, thus to the rainfall pattern [30, 31]. During the first part of the rainy season, sparse rainfall events fill ponds that dry a few days later: this succession of filling and emptying stages is favourable for Ae. vexans vectors [30]. The second half of the rainy season is characterised by frequent and heavy rainfall events: ponds remain flooded for 2-3 months, allowing the *Culex* population to explode. Rainy season, characterised by high ruminant and RVF vector densities, is thus highly favourable for RVFV transmission. However, during the dry period vectors are absent [30], and the mechanisms allowing RVFV recurrence in this region still remain unknown. As demonstrated in other contexts, movements of viraemic ruminants are likely to contribute to RVFV persistence and spread [32-34]. Belkhiria et al. identified three types of migrations in a recent survey performed in the Ferlo: predominant long-distance country level migrations, short distance migrations limited to the Ferlo region with breeders moving from pond to pond, and transboundary migrations that extend from Mauritania to Gambia through Senegal [28]. Whatever the distance they travel, these herds may pass through infected areas and may transport and introduce the virus into the Ferlo. Mauritania is a well-known endemic country, and serological and/or virological evidences of RVFV circulation have been reported, either in south Senegal [35], or in Gambia, where mosquito populations are active yearlong. The second, and still main hypothesis in the literature for persistence of RVFV in the environment between epizootics is vertical transmission in mosquito vectors (VT), *i.e.* the transmission of the virus from infected females to mosquito offspring [36]. This VT mechanism has been demonstrated in Kenya for Aedes mcintoshi (Huang), and a recent survey carried on in Sudan suggested its existence in Culex quinquefasciatus (Say). VT may be an alternative, and non-exclusive, way for RVFV to survive the dry season, and persist in the area despite unfavourable conditions. Although VT has never been demonstrated for Ae. vexans in northern Senegal, and although a recent entomological survey showed very low densities of Ae. mcintoshi in this area [37, 38], the existence of this mechanism cannot be ruled out. Another mechanism could be the aestivation of RVFV-infected mosquitoes able to survive the dry season, similar to the way over-wintering *Culex* are capable of maintaining West-Nile virus in the United States [39–41]. However, laboratory and field arguments are lacking and further studies are needed to bring more substantial evidence of these mechanisms. A final putative mechanism for persistence of RVFV could be through wild reservoirs [42]. However, the association between rodents and the maintenance of RVFV remains controversial [43]. RT-PCR positive results have been observed in Dorcas gazelles (Gazella dorcas) from the fenced natural reserve of Guembeul [35]. However, despite significant effort of Senegal authorities to re-introduce wild ungulate species, populations have largely declined in Senegal, because of habitat loss, overhunting and other human-related onslaughts. African buffalo (Syncerus caffer), the best candidate among wild ungulates for RVFV maintenance in Southern Africa [43], is not present in Senegal, and the densities of other wild ungulates, are so low, that they probably could not sustain a longterm RVFV circulation. Lastly, and because of the absence of mosquitoes during the dry season in the study area, it is unlikely that wild animals contribute to maintain the virus during this unfavourable season [42].

In the Ferlo, the set composed of a temporary pond, associated vectors, sedentary ruminants living around, and nomadic herds seasonally settling around, constitutes the elementary unit of the RVF epidemiological system. Rainfall variation is the main driver for pond surface fluctuations which affects both the vector population dynamics and nomadic migration patterns. In addition, resident and nomadic herd immunity impact the ability of RVFV to circulate. A better understanding this complex system is needed to improve our comprehension of RVF epidemiology in this region, to quantify the main determinants of RVF transmission and emergence and help establishing better surveillance, prevention and control strategies.

The aim of this work was to model RVFV transmission in this epidemiological system, and to use this model to infer the respective contribution of nomadic movements and VT in RVFV recurrence. We first carried out in 2016 epidemiological surveys to (i) document the demographic characteristics of both sedentary and nomadic ruminant populations, as well as the duration of nomadic herds stays in our study area, and (ii) estimate the RVFV seroprevalence rate in sedentary small ruminants and nomadic herds transiting through the study area. Then we modelled a realistic epidemiological system at the pond level integrating both Cx. poicilipes and Ae. vexans population dynamics, resident and nomadic ruminant herd population dynamics, and nomadic herd movements. Vector population dynamics were based on outputs of an entomological model (EM) previously and independently developed, parameterized and validated using mosquito trapping data collected in the same area, during the same period [44]. Model comparison methods allowed analysing the scenarios that could explain the recurrent circulation of the virus in this ecosystem, incorporating VT in Ae. vexans or not, direct transmission between hosts or not, and occasional or regular viral introduction through nomadic herd seasonal migrations. The selected scenario was finally used to estimate the transmission parameters based on the observed seroprevalence data, and provide new insights on the role of nomadic herds in RVFV recurrence in the survey area.

Materials and methods

Ethical statement

No human experiments were conducted in this study. Meetings were organized with Younoufere villagers and nomadic and resident farmers to explain the goals of the study and the decision to participate was taken at the individual level. Informed consent was given verbally and documented in questionnaires. For cultural reasons, written consent could not be obtained. The LNERV, which conducted the fieldwork, has the statutory authority to carry out serological surveys on farm animals. No ethical validation structure for field protocols exists in Senegal. During this study, we followed the World Animal Health Organisation (OIE) guiding principles on animal welfare included in the OIE terrestrial Code, Chapter 7.8 "Use of Animals in research and education" [45].

Study area and epidemiological system

The study was conducted in a 86 sq. km area located near Younoufere village (Fig 1). Located in the Ferlo area, northern Senegal, Younoufere (15.269464° N and 14.463094° W) is 80 km far from Barkedji village, where many epidemiological and/or entomological studies on RVF previously occurred [23, 27, 46–48]. Both areas share the same eco-climatic conditions previously described, i.e. a semi-arid steppe and many temporary ponds filled by seasonal rainfall, and that constitute a perfect habitat for *Cx. poicilipes* and *Ae. vexans* mosquitoes [23, 28, 31, 38].



Fig 1. Map of the study area. Map created using ESRI ArcGIS; land cover map derived from a SPOT-5 image (2014-08-07); administrative boundaries extracted from DIVA-GIS (https://www.diva-gis.org/).

https://doi.org/10.1371/journal.pntd.0008009.g001

A field systematic census performed during the 2015 rainy season indicated the existence of 25 ponds in the study area, corresponding to an average density of 0.29 pond/sq. km. Younoufere is also a known hub for nomadic herds that may come from the Ferlo area itself, but also from northern or southern regions of Senegal [28]. Upon arrival, nomads select a convenient place to settle, never far from a pond, i.e. 100 m to 3 kms [23]. The location of resident herds is also strongly constrained by the availability of livestock watering places. During the rainy season, herds remain located in the neighbourhood of the temporary ponds where sedentary breeders live. During the dry season, sedentary herds have to move every day to distant drilling water place they use as water points.

Epidemiological surveys

Host population sizes, demographic characteristics/parameters, and serological status of resident and nomadic animals were investigated thanks to field surveys described hereafter.

Resident herds population and serological status. There is no existing census of the cattle and small ruminants (i.e. goats and sheep) living in the study area. The average size of sedentary populations was estimated based on the number of animals visiting Younoufere's well during the dry season. Indeed, temporary ponds are dried out during the dry season and farmers are forced to take their animals to the well every day. Local authorities control access of animals to this watering place, and animal owners that come for watering ruminants are systematically registered. In 2012 dry season-the most recent available registration, an estimate of 6000 cattle and 30,000 small ruminants visited the Younoufere well. Eighty percent of these ruminants, i.e 4800 cattle and 24,000 small ruminants, belonged to resident breeders. We assumed that homes of these resident herders were located near temporary ponds (dried during the dry season) within a 15 km radius of the well. Considering a density of 0.29 pond per sq. km (the average pond density in the study area–see above), the expected number of temporary ponds in this 15 km radius area was 205. The average number of sedentary animals living around a given pond was then estimated at 23 cattle (i.e. 4800/205) and 117 small ruminants (i.e. 24000/205).

In May 2016, to assess the serological status of resident herds living in the study area, a cross sectional serological survey was performed in 6 resident herds living in 3 small villages located in the close vicinity of Younoufere (Kodediare, Nacara and Soringo (Fig 1)). Animals were randomly chosen among herds whose owner agreed to participate to the study. As a whole, an age-structured sample of 222 small ruminants were blood sampled (no sample from resident cattle could be obtained). Sera were tested by IgG ELISA test using a Kit ID screen Rift Valley Fever Competition Multi-species from IDvet (Innovative Diagnostics) according to the manufacturer's instructions. Positive samples were then tested for IgM using the Kit ID screen Rift Valley Fever IgM capture. Serological analyses were performed in the virological laboratory of ISRA (Dakar, Senegal).

Nomadic herds population, movements and serological status. The average size of nomadic populations during the dry season was estimated using the data collected at the You-noufere well at 6 cattle and 29 small ruminants in 2012. To estimate the size of nomadic population during the rainy season, we used data from a survey ran between from August 2015 to January 2016 [28]. Briefly, the survey consisted of 6 one week field sessions (4 during the rainy season and 2 in the early dry season) aiming to characterize nomadic movements and their determinants, as well as the RVF serological status of animals when arriving in the study area. The protocol is fully described in [28]. Only one herd was encountered during the last session performed in late January, and this herd could not be included in the sampling frame described below. Session 6 data was then merged with session 5 (Table 1). During each session, the majority of breeders settling in the study area agreed to participate to the study and were interviewed.

Information about their arrival and scheduled departure dates were collected [28], which allowed estimating the duration of their stays around a given temporary pond. Herders were asked to list all locations where they stayed during the past year as well as all the areas they plan to use during the coming months. For each herd the home location (where the herd usually stayed during the dry season) and the locations visited by the herd during the rainy season were identified. As described by Belkhiria et al.[28], the locations were grouped into three regions: the Northern region (including the Senegal river valley), the Ferlo and the Southern region. This allowed to distinguish two types of nomadic breeders: short-range nomads who stayed yearlong in the Ferlo, moving between temporary ponds during the rainy season, and

Session	Period	Season	Number of Breeders	Total number of animals			
				Cattle	Sheep	Goats	
1	15/08/2015 to 25/08/2015	Rainy	37	2591	5433	1268	
2	31/08/2015 to 04/09/2015	Rainy	32	1801	4906	1467	
3	15/09/2015 to 17/09/2015	Rainy	39	1589	4152	1466	
4	01/12/2015 to 04/12/2015	Rainy	14	714	2940	669	
5	09/01/2016 to 11/01/2016	Early dry	9	640	2035	370	
	Average values	Rainy	31	1674	4358	1217	

Table 1. Surveys conducted in nomadic breeders of the study area.

https://doi.org/10.1371/journal.pntd.0008009.t001

the long-range nomads, whose home locations were outside the Ferlo (i.e. in Northern or Southern regions) and who visited the Ferlo during the rainy season, moving likewise between temporary ponds. To assess the serological status of nomadic animals entering the study area, 4 herds newly arrived were selected based on owner willingness during each of the 6 sessions. Among the 132 nomadic herders who accepted to answer the above-described movement survey, 22 accepted to participate in a serosurvey and 30 animals (cattle and small ruminants) were randomly sampled from each. Sera were tested for RVF IgG antibodies using the test mentioned above. Positive sera were tested for IgM.

Statistical analyses

No cattle could be sampled in resident herds and cattle sera were sometimes missing in nomadic herds (Table 2). We thus analyzed the relationship between seropositivity and potential risk factors in small ruminants only, using a logistic mixed model. The outcome was the individual IgG status, and the fixed effects were firstly individual variables: age (years), species (sheep of goat) and sex (male or female).

Secondly, resident herds were sampled in one session, in May 2016, whereas nomadic herds were sampled in 5 sessions, between August 2015 and January 2016. The respective effects of the sampling date and of the herd type (short vs long range nomads), could thus not be disentangled. For that reason, we created a composite variable to represent the joint effect of sampling date and herd type. We first defined three groups of sampling dates: (i) the 2015's rainy season 1st half (i.e. August-September 2015, corresponding to sessions 1–3 in nomadic herds: see Table 1), (ii) the 2015 rainy season's 2nd half and the early 2016 dry season (i.e. October 2015-January 2016, corresponding to sessions 4–5 in nomadic herds: see Table 1), and (iii) during the 2016 dry season (i.e. May 2016, the sampling period for resident herds). We then defined a composite variable with 5 distinct modalities: short-range (resp. long-range) nomadic herds sampled during the 1st half of the 2015 rainy season (August-September 2015),

Sampling period	Season	Type of breeder	Sheep	Goats	Cattle
15/08/2015 - 25/08/2015	Rainy (1 st half)	Short-range nom. ^a	1/26 ^b [0/0 ^{c,d}]	0/13	0/21
		Long-range nom.	0/40	0/15	0/5
31/08/2015 - 04/09/2015	Rainy (1 st half)	Short-range nom.	3/32 [0/3]	5/18 [0/5]	3/10 [0/3]
		Long-range nom.	4/45 [0/4]	1/14 [0/1]	0/1
15/09/2015 - 17/09/2015	Rainy (1 st half)	Short-range nom.	6/31 [3/6]	4/29 [0/3 ^d]	
		Long-range nom.	3/66 [0/3]	2/24 [1/2]	
01/12/2015 - 04/12/2015	Rainy (2 nd half)	Short-range nom.	5/26 [0/5]	1/13 [1/1]	4/21 [1/4]
		Long-range nom.	14/69 [2/14]	2/9 [0/2]	3/12 [0/3]
09/01/2016 - 11/01/2016	Early dry	Short-range nom.	4/64 [1/4]	8/26 [3/8]	
		Long-range nom.	9/30 [3/9]		
May 2016	Dry	Resident	24/168 [0/24]	10/54 [0/10]	
Total		Resident	24/168 (0.14 ^e)	10/54 (0.19)	
		Short-range nom.	19/179 (0.11)	18/99 (0.18)	7/52 (0.13)
		Long-range nom.	30/250 (0.12)	5/62 (0.08)	3/18 (0.17)

Table 2. Results of serosurveys conducted in ruminants of resident and nomadic herds in Younoufere area, Ferlo, Senegal, May 2015-January 2016.

^a Nomads.

^b IgG-positive/tested.

^d Because of the lack of serum, one IgG-positive serum could not be tested for IgM.

^e Proportion of IgG-positive results

https://doi.org/10.1371/journal.pntd.0008009.t002

^c IgM-positive/tested (all IgG-positive sera were tested for IgM, except 2).

or during the 2nd half of the 2015 rainy season and the early 2016 dry season (October 2015-January 2016), and resident herd sampled during the 2016 dry season (May 2016). This composite variable was included in the model as a fixed effect. The breeder was treated as a random effect.

We computed 95% confidence intervals for prevalences and odds-ratios, indicated using squared brackets. All the statistical analyses were conducted using R 3.6.1 [49].

Epidemiological model

Model design and parameterization. The epidemiological system was considered at the pond scale. Considering that nomadic herders always settle close to ponds and that the flight capacities of both Culex and Aedes are highly variable, ranging from 50m to 50km depending on species, climatic and environmental conditions, host availability and experimental protocols [50], we assumed that mosquitoes emerge from ponds to seek hosts in a 2.5 km radius, corresponding to the putative area of influence of the pond in this region. We considered two closed vector populations (*i.e.* without dispersion from or to the epidemiological system): a Cx. poicilipes and an Ae. vexans population. Four host populations were included: cattle and small ruminants from sedentary and nomadic herds. Resident ruminant populations were assumed closed for renewal (no purchase of live animals, culled animals being replaced by animals born in the epidemiological system). However, nomadic ruminant populations were assumed to have a permanent renewal due to arrival of some breeders and the departure of others. Each vector population was represented by a compartmental S-L-I model (S: susceptible vectors, L: infected but non-infectious vectors -during the extrinsic incubation period, I: infectious vectors that may transmit RVFV to the susceptible hosts upon which they feed), and by an additional state variable (G) representing the proportion of infected eggs. Host populations were represented by a S-I-R model (S: susceptible animals, I: viraemic animals, which may transmit RVFV to vectors feeding upon them, R: immune animals), further stratified by age (with yearly age classes) and by sex/physiological status (males, empty females and gestating females).

Model dynamics represented three distinct processes (Fig 2): the population dynamics (birth and death of vectors and hosts), the infection dynamics (RVFV transmission) and the



Fig 2. Schematic representation of the epidemiological model.

https://doi.org/10.1371/journal.pntd.0008009.g002

population renewal (arrival and departure of nomadic breeders). The population dynamics of vectors was not explicitly represented in the model. Instead, we used the outputs (computed with a daily time step) of the entomological model (EM) elaborated by Tran et al., and parameterized using mosquito trapping data collected in the same study area and period (Fig 2) [44]. This model reproduced the demographic dynamics of *Ae. vexans* and *Cx. poicilipes* mosquitoes around a temporary pond, based on the evolution of pond surface, rainfall, temperature and humidity.

Two between-host transmission routes were considered: vector-borne, and direct, considering that when viraemic hosts abort (or calve/lamb), the abortion (or calving/lambing) products are infectious, and susceptible animals are exposed to these highly infectious materials (Fig 2). Vector-borne transmission was parameterized by a scaling factor for vector population sizes (ψ), and direct transmission by a transmission parameter (β). In vectors, RVFV transmission from an infected female to its eggs was parameterized by a VT probability (ω).

Since the host and vector population sizes remained constantly high in the modelled system throughout the year (>500 individuals) and because the RVFV circulation level was also expected to be high due to the documented recurrence of RVFV transmission in the area, we chose to implement the model in a deterministic framework, with a discrete daily time step.

A full description of (i) the epidemiological model, (ii) its coupling with the entomological model, and (iii) its parameterization, are provided in <u>S1 Text</u>.

RVFV introduction scenarios and parameter estimation. The model was simulated during an 8 years duration corresponding to the maximal lifespan of hosts. Entomological inputs (vector population dynamics) were obtained from the EM for the 2008–2015 period, using daily rainfall, temperature and humidity data collected from the study area during this period. Two RVFV introduction scenarios were considered: a single introduction due to the arrival of a given number n_{intro} of viraemic nomadic animals during the 2008 rainy season (scenario A), or successive introductions due to the arrival of n_{intro} viraemic animals during each rainy season from 2008 to 2015 (scenario B). In both cases, the viraemic animals were proportionally allocated to cattle and small ruminants' nomadic populations according to the species-specific numbers of susceptible individuals. For both scenarios, the risk of introduction and a consecutive initiation of transmission cycle was assumed to vary throughout a given rainy season, as (i) the geographic origin of nomadic breeders is known to vary throughout the rainy season [28], (ii) the succession in time of Ae. vexans (first part of the rainy season) and Cx. poicilipes populations (second part of the rainy season) [30] may also imply differences in RVFV circulation level, and (iii) a progressive amplification of RVFV circulation in the whole Ferlo area may induce an increased risk of introduction in the late rainy season. Therefore, the total number of viraemic animals entering the area during a given rainy season (n_{intro}) was divided into (i) a proportion p_{early} arriving during the first half of the rainy season, and (ii) $1-p_{early}$ arriving during its second half. Because of the daily time step, the number of viraemic ruminants entering the system at a given day was expected to be low (at least when the value of n_{intro} is lower or comparable to the duration, in days, of the rainy period). To allow a realistic representation of such low numbers of viraemic animals entering the system, the day of arrival of each of the n_{intro} ruminants was randomly chosen at the beginning of each simulation. This method introduced a stochastic component to the model, justifying the use of a Bayesian framework for parameter estimation, instead of likelihood maximization methods more commonly used for deterministic models. We used an Adaptive Population Monte-Carlo Approximate Bayesian Computation method (ABC-APMC) [51], with non-informative priors (assumed to be independent of one another) for each of the 5 estimated parameters:

• the scaling factor for vector population sizes (ψ), prior: U(0–10)

- the direct transmission parameter to hosts (β), prior: U(0–10)
- the probability that an infected *Aedes* female mosquito transmits the virus to its eggs (ω_{aedes} , this probability being considered null for *Culex* females), prior: U(0–0.10) (this proportion was previously estimated at 0.007 [52])
- the total number of viraemic nomadic ruminants introduced during a rainy season (*n_{intro}*), prior: U(0–1500)
- the proportion of viraemic nomadic ruminants arriving during the 1st half of the rainy season (*p_{early}*), prior: U(0–1).

The summary statistics consisted of the observed numbers of seropositive resident small ruminants per age class (6 classes) found when simulating the serosurvey described above (i.e. during the 2016 dry season, with the age-specific numbers of tested animals). The recommended settings were used for the ABC-APMC algorithm: 0.5 for the quantile of the distribution of distances to observed data used to define tolerance thresholds; 0.05 for the stopping criterion based on the renewal proportion of particles in the sequential Monte-Carlo procedure; and a final size of 10,000 particles to build posterior probabilities [51].

The scenarios A and B were compared using a model selection procedure based on random forest classification methods, specifically designed to allow model comparison in an approximate Bayesian computation framework [53]. Using the same method, the need for VT and/or direct transmission between hosts was tested. Five models were compared (Table 3) and the best model was used for parameter estimation. When two models were equivalent, the most parsimonious was selected.

Results

Serological survey in sedentary herds

One hundred-sixty eight resident sheep and fifty four goats were sampled (Tables 2 and 4). The overall IgG seroprevalence rate was 15.3%. [10.8–20.7], with 14.3% [7.4–25.1] in Kodediaré, 10% [4.4–20.1] in Nakara, and 20% [12.9–31.4] in Soringo. The IgG seroprevalence rate in goats was 18.5% [9.2–31.4] and 14.3% [9.4–20.5] in sheep. We did not register any IgM positive sera in resident herds. The seroprevalence rate increased significantly with age, determined based on the dentition of the animals (Table 4) (test for trend in proportions: p<0.005).

Serological survey in nomadic herds and survey in breeders

An average number of 36 nomadic breeders per session were interviewed during the 3 surveys of the rainy season, and 12 breeders during the 2 surveys of the dry season (Table 1). Based on their declarations, nomadic herders would stay on average 16.3 days in Younoufere during the

Model	Introduction scenario	Vertical transmission parameter in Aedes ω_{aedes}	Direct transmission parameter between hosts $meta$
M0	A (2008 only)	Estimated	Estimated
M1	B (2008–2015)	Estimated	Estimated
M2	Best scenario*	0	0
M3	Best scenario*	Estimated	0
M4	Best scenario*	0	Estimated

Table 3. Model selection: definition of the compared models.

 * According to M0-M1 comparison

https://doi.org/10.1371/journal.pntd.0008009.t003

Age	Dentition	IgG Positive/Tested (proportion) [IgM Positive/Tested]			
		Resident herds	Short-range nomads	Long-range nomads	
0-1 year	Milk teeth	1/35 (0.03) [0/1]	2/61 (0.03) [1/2]	6/91 (0.07) [1/6]	
1-2 years	2 teeth	2/36 (0.06) [0/2]	2/39 (0.05) [1/2]	3/56 (0.05) [1/3]	
2-3 years	4 teeth	8/35 (0.23) [0/8]	13/57 (0.23) [4/12 ^a]	0/30	
3-4 years	6 teeth	4/35 (0.11) [0/4]	9/43 (0.21) [0/9]	8/36 (0.22) [0/8]	
4-5 years	8 teeth	10/39 (0.26) [0/10]	7/46 (0.15) [2/7]	9/42 (0.21) [2/9]	
>5 years	>8 teeth	9/42 (0.18) [0/9]	4/32 (0.12) [0/3 ^a]	9/57 (0.16) [2/9]	
Total		34/222 (0.15) [0/34]	37/278 (0.13) [8/35]	35/312 (0.12) [6/35]	

Table 4. Results of an age-structured serosurvey conducted in small ruminants from 6 resident herds and 22 nomadic herds of Younoufere area, Ferlo, Senegal, August 2015-May 2016.

^a Because of the lack of serum, one IgG-positive serum could not be tested for IgM.

https://doi.org/10.1371/journal.pntd.0008009.t004

rainy season (range: 2–122 days), and 9.8 days during the dry season (range: 1–21 days). Herds were essentially composed of sheep with a variable number of cattle goats (and donkeys, which were not blood sampled). Four herds were dominated by cattle and two by goats. The distribution of the size of the 22 sampled herds as well as their compositions are provided in <u>S1 Table</u> and <u>S1 Fig</u>.

A total number of 660 serological test results were available for the statistical analysis in nomadic sheep, goats and cattle (Table 2) among which 590 were from small ruminants and 70 from cattle. The overall estimated IgG seroprevalence rate was 12.4% [10.0–15.2], specifically 14.3% in cattle (n = 70; [7.4–25.1]), 11.4% (n = 429; [8.6–14.9]) in sheep, and 14.3% in goats (n = 161; [9.5–20.9]). A higher seropositivity rate was detected in samples collected during the 2^{nd} half of the rainy season and the early dry season (Dec-Jan) than in samples collected during the 1^{st} half of the rainy season (Aug-Sept), with respectively 18.5% [14.1–23.7] and 8.2% [5.7–11.4] of sera detected IgG positive. The observed IgM seroprevalence was 1% [0.3–2.6] for nomadic herds sampled during the first half of the rainy season against 4% [2.1–7.2] for herd sampled afterwards.

Statistical model

Animal's age was significantly associated with the seropositivity risk (OR: 1.3 [1.1–1.5] for an increase of 1 year of age, Tables 4 and 5). Small ruminants from nomadic herds (both long-and short-range) were more frequently seropositive when sampled during the 2^{nd} half of the rainy season (and the early dry season) than animals from long-range nomadic herds sampled during the 1^{st} half of the rainy season (the reference class). It was also the case for resident herds sampled during the 2016 dry season (Table 5).

Epidemiological model

Based on the posterior probability (>0.999), model M1 (yearly introductions of viraemic animals, scenario B) was first selected as compared to M0 (a single year with introduction of viraemic animals, in 2008) (Table 3). The comparison of model M1 (including both VT in *Aedes* and direct transmission) with model M2 (including none of these mechanisms) led to select model M1 (posterior probability 0.996). Then model M1 was selected against M3 (same as M1, without direct transmission; posterior probability 0.994). Lastly, M1 and M4 appeared equivalent (posterior probability: 0.32): we thus selected the most parsimonious model M4, which included direct transmission but no VT in *Aedes*, and repeated introductions of RVFV by nomadic animals, during each rainy season, from 2008 to 2015.

Variable	Value	p-value	Odds-ratio
Age	Quantitative (years)	< 0.0001	1.3 ^a [1.1–1.5]
Sex	Female	Reference	
	Male	0.14	NS
Species	Sheep	Reference	
	Goat	0.25	NS
Type of herd-	Long-range nom. ^b – 1 st half of rainy season ^c	Reference	
sampling date	Short-range nom.– 1 st half of rainy season ^c	0.09	NS
	Long-range nom.– 2 nd half of rainy season ^c	0.0005	7.9 [2.3–28.3]
	Short-range nom.– 2 nd half of rainy season ^c	0.009	4.5 [1.4–16.2]
	Resident–Dry season ^d	0.02	3.9 [1.2–14.7]

Table 5. Logistic mixed model of the association between seropositivity with individual-level and group-level covariates in small ruminants from resident and nomadic herds of Younoufere area, Ferlo, Senegal, August 2015-May 2016.

^a computed for an increase of 1 year of age

^b Nomads

^c 2015

^d 2016

https://doi.org/10.1371/journal.pntd.0008009.t005

Model fit obtained after the parameter estimation of model M4 appeared satisfactory (Fig 3): the predicted age-specific distributions of IgG seroprevalence rate matched the results of the serosurvey in sedentary small ruminants, and the 95% confidence intervals of age-specific predicted distributions of seroprevalence rates always included the observed value (Fig 3).



Age class

Fig 3. Predicted distribution of seroprevalence rate by age class in sedentary small ruminants. Model is model M4 which combines direct transmission between hosts but no VT in Aedes, and repeated introductions of RVFV by viraemic nomadic animals, each rainy season, from 2008 to 2015. Stars: observed values, blue dashes: 2.5% and 97.5% percentiles of predicted distributions.

https://doi.org/10.1371/journal.pntd.0008009.g003



Fig 4. Posterior distributions of estimated parameters. Model is M4 which combines direct transmission between hosts but no VT in *Aedes*, and repeated introductions of RVFV by viraemic nomadic animals, each rainy season, from 2008 to 2015. ψ : scaling factor for vector population sizes, β : direct transmission parameter between hosts, n_{intro} : total number of viraemic nomadic animals introduced during a rainy season, p_{intro} : proportion of those animals arriving during the 1st half of the rainy season, p_{early} : proportion of viraemic nomadic ruminants arriving during the 1st half of the rainy season.

https://doi.org/10.1371/journal.pntd.0008009.g004

The posterior distributions of estimated parameters are provided in Fig 4. The posterior distribution of the scaling factor for vector population sizes (ψ) had a median value of 0.09 (95% credibility intervals [CI]: 0.005–0.32). The median value of the direct transmission parameter (β), was 4.8 (95% CI: 1.8–9.3). This estimation is close to the estimate previously obtained in Madagascar highlands, by Nicolas et al. (3.2 [3.0–3.4]). The posterior median value of the total number of viraemic nomadic animals introduced during a rainy season (n_{intro}) was 617 (95% CI: 211–1,336). The posterior median of the proportion of viraemic animals arriving during the 1st half of the rainy season (p_{early}) was 0.31 (95% CI: 0.02–0.88): according to the model, viraemic animals were more likely introduced during the late rainy season than early, which is consistent with a progressive amplification of RVFV in the area.

More precisely, combining the two preceding posterior distributions resulted in a median number of 172 (95% CI: 8-742) and 390 (95% CI: 48-1,097) viraemic animals introduced during the 1st and 2nd halves of the rainy season, respectively. Moreover, a total number of 22,686 nomadic animals (3,796 cattle and 18,890 small ruminants) entered the study area during the 2015 rainy season, of which 50% (11,343 animals) arrived in the first half of the rainy season and 50% during the second half of the rainy season. The predicted proportions of viraemic animals among nomadic ruminants entering the study area in 2015 had then a median value of 1.5% [0.1-6.5] for the 1st half of the rainy season and 3.4% [0.4-9.7] for the 2nd half. These predicted proportions can be compared with the observed IgM seroprevalence rates. Indeed, according to Pépin et al. [54], in half of infected animals, the duration of viraemia is about 7 days while the IgM response remains detectable for about 1 month. Viremia duration thus approximately corresponds to 25% of the duration of the IgM response, and one in four IgM positive animals would therefore be viraemic. The observed IgM seroprevalence of 1% and 4% would then correspond to proportions of respectively 0.25% and 1% of viraemic animals among nomadic ruminants entering the study area, during the 1st and 2nd halves of the 2015 rainy season. Although the point estimates are somewhat higher, both figures are consistent with their predicted counterparts since they are included in the credibility intervals: 1.5% [0.1– 6.5] and 3.4% [0.4-9.7].

Discussion

RVF has been repeatedly reported in northern Senegal since 1987, when the first large outbreak of west Africa started in southern Mauritania [47, 55, 56]. Since then, 5 outbreaks were reported by the national RVF surveillance network in 2003, and a survey showed an active viral circulation with multiple clinical cases in small ruminants in the Ferlo area [23]. A longitudinal serosurvey carried in the same area during the 2004 rainy season confirmed a recurrent RVFV transmission [57]. More recently, RVFV was isolated from Ae ochraceus in the same area thanks to a mosquito-based surveillance implemented since 1990 [48]. The last evidence of RVFV circulation occurred during the 2013-2014 outbreak with the detection of one human case in Linguere town located in northern Senegal [58]. Our serological results confirm a long lasting RVF endemicity in resident herds. As expected, we did not detect IgM because residents animals were sampled during the dry season, although having tested for IgM only IgG-positive sera may have led to an underestimate of IgM seroprevalence [59]. This survey also provides the first estimation of RVF IgG seroprevalence in nomadic herds in West Africa, which was close to what was estimated in resident herds. In small ruminants, whatever the type of herd (resident, short-range nomad and long-range nomad), seroprevalence data increases with age. Despite the small sample size, these results suggest that RVF force of infection is equally high in both nomadic and resident herds because of an endemic circulation of RVFV in the whole Ferlo region and a consecutive regular exposition of herds to the virus. In addition, the multivariate analysis (logistic mixed model), showed that, besides the increase of seropositivity risk with age, the seropositivity risk increased between the 1st and 2nd half of the rainy season. This suggests an amplification of the transmission cycle during the rainy season with a peak of circulation at the end of that season. This latter assumption is reinforced by an observed IgM seroprevalence of 1% for nomadic herds sampled during the 1st half of the rainy season, against 4% for herds sampled on late rainy season, and the isolation of a single RVFV strain in the same area in September 2012 (2^{nd} half of the rainy season [48]). The observed IgM rates show that RVFV has been circulating within the Ferlo area between August 2015 and December 2016, and, similarly to what has already been observed in many endemic countries [60–62], without any reported clinical case, either in ruminants nor humans. Lastly the

observed IgM rate is in line with previous results reported in other endemic areas, i.e. 0.5% in Tanzania, 0–4% in Mayotte and 1.3% recently in South Africa [60, 63, 64].

Given the absence of mosquito vectors during the dry season, the way the virus is maintained in the area remains unknown. Here, we assumed that RVF transmission recurrence may be explained by a source-sink dynamic, at least at the Ferlo scale, nomadic herds constantly moving from ponds to ponds according to the availability of water and grass, thus introducing occasionally or seasonally the virus in epidemiological systems where the absence of mosquitoes during the dry season prevents a yearlong virus transmission. Depending on resident herd immunity, the level of which may vary according to the proportion of small ruminants within herds (as the life expectancy is lower in small ruminants than in cattle), the virus does not probably circulate every year around each pond and with the same intensity. However, nomadic herd movement epidemiologically link ponds to each other, and allow the observed circulation recurrence. To investigate this hypothesis, we modelled RVF transmission at the pond scale, during an 8 year time period (corresponding to the average life span of ruminants), incorporating both Ae. vexans and Cx. poicilipes population dynamics, resident and nomadic ruminant population dynamics and movements. Five models were compared that combined a single or yearly introductions of RVFV with the presence or absence of VT and of direct transmission. The selected model incorporated yearly introductions of RVFV from 2008 to 2015 (the study period) by nomadic herds, with a larger proportion of viraemic animals arriving during the 2nd half of the rainy season. This result is consistent with the assumption of an amplification of transmission during the rainy season. The predicted proportion of viremic animals introduced during the 2015 rainy season is also in line with the IgM rate observed in nomadic herds, sampled during either the first (1%) or the second half of the rainy season (4%). In addition, since IgM antibodies do not persist beyond the 50th day after infection in the majority of cases [54], and since the majority of IgM positive animals were sampled in December-January, the infection of these animals likely occurred in November-December, thus confirming the amplification of RVFV circulation at the end of the rainy season.

Implication of vertical transmission (VT) in RVFV persistence in a given area during interepizootic periods remains widely cited. In our study, the selected model did not incorporate VT mechanisms, suggesting that (i) VT is not necessary to account for RVFV recurrence in the Ferlo area, and (ii) nomadic herd movements from areas where the virus circulates yearround (e.g. the Senegal River valley) are sufficient to explain the reactivation of the transmission cycle in the Ferlo each year, while nomadic herd movements within the Ferlo lead to an amplification of the transmission cycle during the rainy season. As a matter of fact, the implication of VT in the persistence of the virus in the field during unfavourable seasons, only relies on few observations. First, RVFV was detected in females and males Ae. mcintoshi, reared in laboratory from field collected eggs in Kenya [37]. However, and to our knowledge, these findings have never been reported again, even during epizootic periods, and whatever the geographic area. Viral RNA was further detected in adult male Ae. vexans and Cx. quinquefasciatus in Sudan in 2009–2010. However, the presence of infectious particles has never been confirmed [65]. The last argument relies on the fact that several outbreaks occurred at the same time in regions separated by hundreds of kilometres, but concurrently exposed to heavy floodings that may have triggered a massive Aedes population hatching, some of these Aedes being possibly infectious [36]. However, such heavy flooding also attract nomadic herds and provoke mosquito population explosions, both favourable to RVFV spread, transmission and animal or human RVF case occurrence after several years of a low-level virus circulation.

The selected model included a direct transmission mechanism between hosts to take into account the excretion of RVFV in abortion products (or lambing/calving products) of viraemic

females, that may be a source of infection for other ruminants (as they are for humans). The estimate of the direct transmission parameter was close to the value obtained by Nicolas et al [34] in a very different ecosystem (Madagascar highlands). Although limited to specific conditions (abortion or calving/lambing of viraemic animals), this direct transmission mechanism may thus play a significant role in the local RVFV circulation, while being independent of the eco-climatic context.

RVFV was recently isolated from Ae. ochraceus [48], and several entomological surveys showed the presence of many potential RVFV vectors such as Mansonia uniformis or Cx. tritaenhyorynchus in the Ferlo. However Ae. vexans and Cx. poicilipes are considered the main vectors of RVF in this region [24, 29]. Their respective population dynamics was first described by Mondet et al in 2002 [30], with an early rainy season characterized by an abundant population of Ae. vexans, whereas the second part of the rainy season was favorable to Culex population. These observations have been recently confirmed [38], with Ae. vexans arabiensis representing 94.98% of mosquitoes captured, and Cx. poicilipes 0.24% of the mosquito trapped. Despite this low proportion of *Cx. poicilipes*, we chose to incorporate both vector species in our model, to account for seroconversions occurring late in the rainy season, when Ae. vexans is absent. Field studies conducted in the study area suggest that the flight distance of these two mosquito species is relatively low (less than 1500m according to [66]). However, the ability to perform long-distance movements, as demonstrated in Anopheles [67], could allow mosquitoes avoiding the dry season and re-colonize the Ferlo area at the beginning of each rainy season. This adaptive strategy has not been described in either Culex or Aedes, but it cannot be excluded and could contribute to the introduction of the virus into the Ferlo, and to its circulation within the Ferlo. If it is the case, this process could not be distinguished from the movements of nomadic herds in our mathematical model.

The number and size of nomadic (and sedentary) herds were assumed constant during the study period, although the duration of the rainy season varies from year to year, which may increase or decrease the number and size of nomadic herds. We made the hypothesis that the duration of the 2015 rainy season was not very different from the average, as confirmed by the estimated dates for the beginning and end of the 2008–2014 rainy seasons (Table F in <u>S1 Text</u>), and that the ruminant population dynamics parameters observed in 2015 were close to the usual conditions prevailing in the study area.

Outbreaks of vector-borne diseases such as RVF are known to be sensitive to both host movements and landscape characteristics, with factors such as host densities and movement patterns contributing to disease maintenance [34, 68, 69]. RVF is endemic in the Ferlo area. In line with Favier et al [70] who suggested that (i) vertical transmission or wild reservoirs are not necessary to explain RVFV endemicity in the Ferlo, (ii) herd movements in some specific environment can allow endemicity at a regional scale while circulation is epidemic at a local scale, we quantitatively demonstrate in this work that although the existence of a vertical transmission mechanism in *Aedes* cannot be ruled out, nomadic movements are sufficient to account for this endemic circulation in the Ferlo area.

A next outbreak will inevitably occur in the Ferlo area, and the increased national and transboundary nomadic and commercial movements of ruminant herds may allow the virus spreading over large distances, threatening disease-free areas. It is thus urgent to improve the surveillance capacity as well as our knowledge on nomadic herds, their movements and the determinants of these movements. Since they should be targeted in any vaccination campaign, it is finally necessary to learn more about the perception of RVF by nomadic breeders, and the social acceptability of prevention, surveillance, and control measures, such as cattle, small ruminant and dromedary vaccination to protect people, or animal movement restrictions to avoid RVFV spread through hubs in the livestock-trade networks

Supporting information

S1 Checklist. STROBE checklist of items that should be included in reports of observational studies. (DOCX)

S1 Table. Characteristics of the nomadic herds surveyed. (DOCX)

S1 Fig. Distribution of the size (total number of animals) of the surveyed herds (each bar represents a herd; session 1: 21-24/08/2015, session 2: 3-4/09/2015, session 3: 17-18/09/2015, session 4: 1-4/12/2015, session 5: 09-11/01/2016). (TIF)

S1 Text. Epidemiological model. (DOCX)

Acknowledgments

We gratefully thank Younoufere authorities and breeders.

Author Contributions

Conceptualization: Benoit Durand, Annelise Tran, Jaber Belkhiria, Véronique Chevalier.

Data curation: Benoit Durand, Annelise Tran, Jaber Belkhiria, Vladimir Grosbois, Véronique Chevalier.

Formal analysis: Benoit Durand, Vladimir Grosbois, Véronique Chevalier.

Funding acquisition: Véronique Chevalier.

- **Investigation:** Moustapha Lo Modou, Annelise Tran, Aminata Ba, Fafa Sow, Jaber Belkhiria, Assane Gueye Fall, Biram Biteye, Véronique Chevalier.
- Methodology: Benoit Durand, Moustapha Lo Modou, Annelise Tran, Jaber Belkhiria, Assane Gueye Fall, Vladimir Grosbois, Véronique Chevalier.
- Project administration: Véronique Chevalier.

Resources: Véronique Chevalier.

Software: Benoit Durand, Annelise Tran.

Supervision: Moustapha Lo Modou, Annelise Tran, Fafa Sow, Assane Gueye Fall, Véronique Chevalier.

Validation: Benoit Durand, Véronique Chevalier.

Visualization: Benoit Durand, Annelise Tran, Jaber Belkhiria.

Writing - original draft: Benoit Durand, Véronique Chevalier.

Writing – review & editing: Benoit Durand, Moustapha Lo Modou, Jaber Belkhiria, Assane Gueye Fall, Véronique Chevalier.

References

1. Parham P, Waldock J, Christophides G, Hemming D, Agusto F, Evans K, et al. Climate, environmental and socio-economic change: weighing up the balance in vector-borne disease transmission. Philos Trans R Soc B Biol Sci. 2015; 370.

- Kilpatrick A, Randolph S. Drivers, dynamics, and control of emerging vector-borne zoonotic diseases. Lancet. 2012; 380:1946–55. https://doi.org/10.1016/S0140-6736(12)61151-9 PMID: 23200503
- Daubney R, Hudson J, Garnham P. Enzootic hepatitis or Rift Valley fever: an undescribed disease of sheep, cattle and man from east Africa. Journal of Pathology and Bacteriology. 1931; 89:545–79.
- Peyre M, Chevalier V, Abdo-Salem S, Velthuis A, Antoine-Moussiaux N, Thiry E, et al. A Systematic Scoping Study of the Socio-Economic Impact of Rift Valley Fever: Research Gaps and Needs. Zoonoses Public Health 2014; https://doi.org/10.1111/zph.12153 PMID: 25256804
- Swanepoel R, Coetzer JAW. Rift valley fever. In: Coetzer JA, Tustin R. C Editors, second edition, editor. Infectious Diseases of livestock. Cape Town: Oxford university Press, Southern Africa; 2004. p. 1037–70.
- 6. Wilson ML. Rift Valley fever virus ecology and the epidemiology of disease emergence. Ann NY Acad Sci. 1994; 740:169–80. https://doi.org/10.1111/j.1749-6632.1994.tb19867.x PMID: 7840448.
- Linthicum K, Britch S, Anyamba A. Rift Valley fever: An emerging mosquito borne disease. Annu Rev Entomol. 2016; 61:395–415. https://doi.org/10.1146/annurev-ento-010715-023819 PMID: 26982443
- Nicolas G, Durand B, Tojofaniiry T, Lacote S, Chevalier V, Marianneau P. A three years serological and virological cattle follow-up in Madagascar highlands suggests a non-classical transmission route of Rift Valley Fever virus. Am J Trop Med Hyg. 2013; 90(2):265–6. <u>https://doi.org/10.4269/ajtmh.13-0538</u> PMID: 24366500
- Olive M, Chevalier V, Grosbois V, Tran A, Andriamandimby S, Durand B, et al. Integrated analysis of environment, cattle and human serological data: risks and mechanisms of transmission of Rift Valley fever in Madagascar. PLoS Negl Trop Dis. 2016; 10(7). <u>https://doi.org/10.1371/journal.pntd.0004827</u> PMID: 27415438
- Bob N, Bâ H, Fall G, Ishagh E, Diallo M, Sow A, et al. Detection of the Northeastern African Rift Valley Fever Virus Lineage During the 2015 Outbreak in Mauritania. Open Forum Infect Dis. 2017; 4(2):: ofx087. https://doi.org/10.1093/ofid/ofx087 eCollection 2017 Spring. PMID: 28638845
- Redding D, Tiedt S, Lo Iacono G, Bett B, Jones K. Spatial, seasonal and climatic predictive models of Rift Valley fever disease across Africa. Philos Trans R Soc Lond B Biol Sci. 2017; 372(1725):pii: 20160165. https://doi.org/10.1098/rstb.2016.0165 PMID: 28584173
- Rissmann M, Eiden M, Wade A, Poueme R, Abdoulkadiri S, Unger H, et al. Evidence for enzootic circulation of Rift Valley fever virus among livestock in Cameroon. Acta Trop. 2017; 127:7–13. <u>https://doi.org/10.1016/j.actatropica.2017.04.001</u> Epub 2017 Apr 17. PMID: 28427961
- Fafetine J, Coetzee P, Mubemba B, Nhambirre O, Neves L, Coetzer J, et al. Rift Valley Fever Outbreak in Livestock, Mozambique, 2014. Emerg Infect Dis. 2016; 22(12):2165–7. <u>https://doi.org/10.3201/</u> eid2212.160310 PMID: 27869589
- Murithi R, Munyua P, Ithondeka P, Macharia J, Hightower A, Luman E, et al. Rift Valley fever in Kenya: history of epizootics and identification of vulnerable districts. Epidemiol Infect. 2010; 139:372–80. https://doi.org/10.1017/S0950268810001020 PMID: 20478084.
- Mroz C, Gwida M, El-Ashker M, El-Diasty M, El-Beskawy M, Ziegler U, et al. Seroprevalence of Rift Valley fever virus in livestock during inter-epidemic period in Egypt, 2014/15. BMC Vet Res 2017; 13(1):87. https://doi.org/10.1186/s12917-017-0993-8 PMID: 28381251
- Tambo E, Olalubi O, Sacko M. Rift valley fever epidemic in Niger near border with Mali. Lancet Infect Dis. 2016; 16(12):1319–20. https://doi.org/10.1016/S1473-3099(16)30477-7 Epub 2016 Nov 15. PMID: 27998581
- Métras R, Fournié G, Dommergues L, Camacho A, Cavalerie L, Mérot P, et al. Drivers for Rift Valley fever emergence in Mayotte: A Bayesian modelling approach. PLoS Negl Trop Dis. 2017; 11(7):: e0005767. https://doi.org/10.1371/journal.pntd.0005767 eCollection 2017 Jul. PMID: 28732006
- Olive M, Grosbois V, Tran A, Nomenjanahary L, Rakotoarinoro M, Andriamandimby S, et al. Reconstruction of Rift Valley fever transmission dynamics in Madagascar: estimation of force of infection from seroprevalence surveys using Bayesian modelling. Sci Rep. 2017; 7:39870. <u>https://doi.org/10.1038/</u> srep39870 PMID: 28051125
- Memish Z, Masri M, Anderson B, Heil G, Merrill H, Khan S, et al. Elevated antibodies against Rift Valley fever virus among humans with exposure to ruminants in Saudi Arabia. Am J Trop Med Hyg. 2015; 92 (4):739–43. https://doi.org/10.4269/ajtmh.14-0575 Epub 2015 Feb 2. PMID: 25646253
- Lagare A, Fall G, Ibrahim A, Ousmane S, Sadio B, Abdoulaye M, et al. First occurrence of Rift Valley fever outbreak in Niger, 2016. Vet Med Sci. 2018. https://doi.org/10.1002/vms3.135 [Epub ahead of print]. PMID: 30411868
- Shoemaker T, Nyakarahuka L, Balinandi S, Ojwang J, Tumusiime A, Mulei S, et al. First Laboratory-Confirmed Outbreak of Human and Animal Rift Valley Fever Virus in Uganda in 48 Years. Am J Trop Med Hyg. 2019; 100(3):659–67. https://doi.org/10.4269/ajtmh.18-0732 PMID: 30675833

- Promed. PRO/AH/EDR> Rift Valley fever—Mayotte (04): more cases, human, cattle Promed Mail; 2019 [cited 2019 2019-03-03 12:48:56]. Available from: http://www.isid.org.
- Chevalier V, Lancelot R, Thiongane Y, Sall B, Mondet B. Incidence of Rift Valley fever in small ruminants in the Ferlo pastoral system (Senegal) during the 2003 rainy season. Emerg Inf Dis. 2005; 11 (11):1693–700.
- Diallo M, Lochouarn L, Ba K, Sall AA, Mondo M, Girault L, et al. First isolation of the Rift Valley fever virus from *Culex poicilipes (Diptera: Culicidae)* in nature. Am J Trop Med Hyg. 2000; 62(6):702–4. https://doi.org/10.4269/ajtmh.2000.62.702 PMID: 11304058.
- Marrama L, Spiegel A, Ndiaye K, Sall AA, Gomes E, Diallo M, et al. Domestic transmission of Rift Valley Fever virus in Diawara (Senegal) in 1998. Southeast Asian JTropMedPublic Health. 2005; 36(6):1487– 95.
- Ba Y, Sall A, Diallo D, Mondo M, Girault L, Dia I, et al. Re-emergence of Rift Valley fever virus in Barkedji (Senegal, West Africa) in 2002–2003: identification of new vectors and epidemiological implications. J Am Mosq Control Assoc. 2012; 28(2):170–8.
- Sow A, Faye O, Ba Y, Diallo D, Fall G, Faye O, et al. Widespread Rift Valley Fever Emergence in Senegal in 2013–2014. Open Forum Infect Dis. 2016; 3(3). doi: ofw149 eCollection 2016. PMID: 27704007
- Belkhiria J, Lo M, Sow F, Martínez-López B, Chevalier V. Application of Exponential Random Graph Models to determine Nomadic herders' movements in Senegal. Trans Emerg Dis. 2019. <u>https://doi.org/ 10.1111/tbed.13198</u> PMID: 30959578
- Fontenille D, Traore-Lamizana M, Diallo M, Thonnon J, Digoutte JP, Zeller HG. New vectors of Rift Valley fever in West Africa. Emerg Infect Dis. 1998; 4(2):289–93. <u>https://doi.org/10.3201/eid0402.980218</u> PMID: 9621201.
- 30. Mondet B, Diaïté A, Ndione JA, Fall AG, Chevalier V, Lancelot R, et al. Rainfall patterns and population dynamics of Aedes (Aedimorphus) vexans arabiensis Patton, 1905 (Diptera, Culicidae), a potential vector of Rift Valley fever virus in Senegal. J Vector Ecology. 2005; 30(1):102–10.
- Soti V, Tran A, Degenne P, Chevalier V, Lo Seen D, Thiongane Y, et al. Combining hydrology and mosquito population models to identify the drivers of Rift Valley Fever emergence in semi-arid regions of West Africa. PLoS Negl Trop Dis. 2012; 6(8):e1795. https://doi.org/10.1371/journal.pntd.0001795 PMID: 22928055
- Abdo-Salem S, Waret-Szkuta A, Roger F, Olive MM, Saeed K, Chevalier V. Risk assessment of the introduction of Rift Valley fever from the Horn of Africa to Yemen via legal trade of small ruminants. Tropical Animal Health and Production. 2011; 43(2):471–80. <u>https://doi.org/10.1007/s11250-010-9719-</u> 7 PMID: 20967567
- Lancelot R, Béral M, Rakotoharinome V, Andriamandimby S, Héraud J, Coste C, et al. Drivers of Rift Valley fever epidemics in Madagascar. Proc Natl Acad Sci U S A. 2017; 114(5):938–43. https://doi.org/ 10.1073/pnas.1607948114 Epub 2017 Jan 17. PMID: 28096420
- 34. Nicolas G, Chevalier V, Tantely L, Fontenille D, Durand B. A spatially explicit metapopulation model and cattle trade analysis allow identifying the key determinants of the recurrent circulation of Rift Valley fever virus in a pilot area of Madagascar highlands Plos Neg Trop Dis. 2014; 8(12):e3346.
- Sow A, Faye O, Ba Y, Diallo D, Fall G, Faye O, et al. Widespread Rift Valley Fever Emergence in Senegal in 2013–2014. Open Forum Infect Dis. 2016; 3(3):ofw149. Epub 2016/10/06. <u>https://doi.org/10.</u> 1093/ofid/ofw149 PMID: 27704007; PubMed Central PMCID: PMC5047427.
- Lumley S, Horton D, Hernandez-Triana L, Johnson N, Fooks A, R H. Rift Valley fever virus: strategies for maintenance, survival and vertical transmission in mosquitoes. J Gen Virol. 2017; 98(5):875–87. https://doi.org/10.1099/jgv.0.000765 Epub 2017 May 30. Review. PMID: 28555542
- Linthicum KJ, Davis FG, Kairo A, Bailey C.L. Rift Valley fever virus. Isolations from *Diptera* collected during an inter-epidemic period in Kenya. J Hyg. 1985; 95:197–209. <u>https://doi.org/10.1017/</u> s0022172400062434 PMID: 2862206
- Biteye B, Fall A, Ciss M, Seck M, Apolloni A, Fall M, et al. Ecological distribution and population dynamics of Rift Valley fever virus mosquito vectors (Diptera, Culicidae) in Senegal. Parasit Vectors. 2018; 11 (1):27. https://doi.org/10.1186/s13071-017-2591-9 PMID: 29316967
- Anderson J, Main A, Cheng G, Ferrandino F, Fikrig E. Horizontal and Vertical Transmission of West Nile Virus Genotype NY99 by Culex salinarius and Genotypes NY99 and WN02 by Culex tarsalis. Am J Trop Med Hyg. 2012; 86(1):134–9. https://doi.org/10.4269/ajtmh.2012.11-0473 PMID: 22232464
- Nelm B, Fechter-Leggett E, Carroll B, Macedo P, Kluh S, Reisen W. Experimental and natural vertical transmission of West Nile virus by California Culex (Diptera: Culicidae) mosquitoes. J Med Entomol. 2013; 50(2):371–8. https://doi.org/10.1603/me12264 PMID: 23540126

- Nelms B, Macedo P, Kothera L, Savage H, Reisen W. Overwintering biology of Culex (Diptera: Culicidae) mosquitoes in the Sacramento Valley of California. J Med Entomol. 2013; 50(4):773–90. https://doi.org/10.1603/me12280 PMID: 23926775
- Gora D, Yaya T, Jocelyn T, Didier F, Maoulouth D, Amadou S, et al. The potential role of rodents in the enzootic cycle of Rift Valley fever virus in Senegal. Microbes Infect. 2000; 2(4):343–6. <u>https://doi.org/ 10.1016/s1286-4579(00)00334-8 PMID: 10817634</u>.
- Olive M, Goodman S, Reynes J. The role of wild mammals in the maintenance of Rift Valley fever virus. Journal of Wildlife disease. 2012; 48(2):241–66.
- 44. Tran A, Fall A, Biteye B, Ciss M, Gimonneau G, Castets M, et al. Spatial modelling of mosquito vectors for Rift Valley fever virus in Northern Senegal: integrating satellite-derived meteorological estimates in population dynamics models. Remote Sensing 2019; 11:1024. https://doi.org/10.3390/rs11091024
- 45. OIE. Use of Animals in research and education [Internet]2016 02 March 2017.
- Chevalier V, Thiongane Y, Lancelot R. Endemic transmission of Rift Valley Fever in Senegal. TransboundEmergDis. 2009; 56(9–10):372–4.
- Zeller HG, Fontenille D, Traore-Lamizana M, Thiongane Y, Digoutte JP. Enzootic activity of Rift Valley fever virus in Senegal. Am J Trop Med Hyg. 1997; 56(3):265–72. https://doi.org/10.4269/ajtmh.1997. 56.265 PMID: 9129528.
- Ndiaye E, Diallo D, Fall G, Ba Y, Faye O, Dia I, et al. Arboviruses isolated from the Barkedji mosquitobased surveillance system, 2012–2013. BMC Infect Dis. 2018 18(1):642. <u>https://doi.org/10.1186/</u> s12879-018-3538-2 PMID: 30541472
- **49.** Team RC. R: A language and environment for statistical computing. In: R Foundation for Statistical Computing, editor. Vienna, Austria2018.
- Verdonschot P, Besse-Lototskaya A. Flight distance of mosquitoes (Culicidae): A metadata analysis to support the management of barrier zones around rewetted and newly constructed wetlands. Limnologica. 2014; 45:69–79.
- Lenormand M, Jabot F, Deffuant G. Adaptive approximate Bayesian computation for complex models. Comput Stat. 2013; 28:2777–96. https://doi.org/10.1007/s00180-013-0428-3.
- Linthicum KJ, Davies FG, Kairo A, Bailey CL. Rift Valley fever virus (family *Bunyaviridae*, genus *Phlebovirus*). Isolations from *Diptera* collected during an inter-epizootic period in Kenya. J Hyg. 1985; 95 (1):197–209. https://doi.org/10.1017/s0022172400062434 PMID: 2862206.
- Pudlo P, Marin J, Estoup A, Cornuet J, Gautier M, Robert C. Reliable ABC model choice via random forests. Bioinforma Oxf Engl. 2016; 32:859–66. https://doi.org/10.1093/bioinformatics/btv684.
- Pépin M, Bouloy M, Bird BH, Kemp A, Paweska JT. Rift Valley fever (Bunyaviridae:Phlebovirus): an update on pathogenesis, molecular epidemiology, vectors, diagnostics and prevention. Vet Res. 2010:41–61. https://doi.org/10.1051/vetres/2010013 PMID: 20167202
- 55. Fontenille D, Traore-Lamizana M, Zeller H, Mondo M, Diallo M, Digoutte JP. Short report: Rift Valley fever in western Africa: isolations from Aedes mosquitoes during an interepizootic period. Am J Trop Med Hyg. 1995; 52(5):403–4. https://doi.org/10.4269/ajtmh.1995.52.403 PMID: 7771605.
- Nabeth P, Kane Y, Abdalahi MO, Diallo M, Ndiaye K, Ba K, et al. Rift Valley fever outbreak, Mauritania, 1998: seroepidemiologic, virologic, entomologic, and zoologic investigations. Emerg Infect Dis. 2001; 7 (6):1052–4. https://doi.org/10.3201/eid0706.010627 PMID: 11747742.
- Chevalier V, Thiongane Y, Lancelot R. Endemic transmission of Rift Valley Fever in Senegal. Trans Emerg Dis. 2009; 56(9–10):372–4.
- Sow A, Faye O, Ba Y, Diallo D, Fall G, Faye O, et al. Widespread Rift Valley Fever Emergence in Senegal in 2013–2014. Open Forum Infect Dis. 2016; 3(3):ofw149. https://doi.org/10.1093/ofid/ofw149 PMID: 27704007
- Georges TM, Justin M, Victor M, Marie KJ, Mark R, Leopold MMK. Seroprevalence and Virus Activity of Rift Valley Fever in Cattle in Eastern Region of Democratic Republic of the Congo. J Vet Med. 2018; 2018:4956378. Epub 2018/07/28. https://doi.org/10.1155/2018/4956378 PMID: 30050953; PubMed Central PMCID: PMC6046122.
- 60. van den Bergh C, Venter E, Swanepoel R, Thompson P. High seroconversion rate to Rift Valley fever virus in cattle and goats in far northern KwaZulu-Natal, South Africa, in the absence of reported outbreaks. PLoS Negl Trop Dis. 2019; 13(5):e0007296. https://doi.org/10.1371/journal.pntd.0007296 eCollection 2019 May. PMID: 31050673
- Nanyingi M, Muchemi G, Thumbi S, Ade F, Onyango C, Kiama S, et al. Seroepidemiological Survey of Rift Valley Fever Virus in Ruminants in Garissa, Kenya. Vector Borne Zoonotic Dis. 2017; 17(2):141–6. https://doi.org/10.1089/vbz.2016.1988 PMID: 27929928

- Umuhoza T, Berkvens D, Gafarasi I, Rukelibuga J, Mushonga B, Biryomumaisho S. Seroprevalence of Rift Valley fever in cattle along the Akagera-Nyabarongo rivers, Rwanda. J S Afr Vet Assoc. 2017; 88 (0):e1–e5.
- Sumaye R, Geubbels E, Mbeyela E, Berkvens D. Inter-epidemic transmission of Rift Valley fever in livestock in the Kilombero River Valley, Tanzania: a cross-sectional survey. PLoS Negl Trop Dis. 2013; 7 (8):e2356. https://doi.org/10.1371/journal.pntd.0002356 eCollection 2013. PMID: 23951376
- Cetre-Sossah C, Billecocq A, Lancelot R, Defernez C, Favre J, Bouloy M, et al. Evaluation of a commercial competitive ELISA for the detection of antibodies to Rift Valley fever virus in sera of domestic ruminants in France. PrevVetMed. 2009; 90(1–2):146–9.
- 65. Mohamed R, Albdelgadir D, Bashab H. Transovarian transmission of Rift Valley Fever Virus by two species of mosquitoes in Khartoum State (Sudan): Aedes vexans (Meigen) and Culex quinquefasciatus (Say) Transovarian transmission of Rift Valley Fever Virus by two species of mosquitoes in Khartoum State (Sudan): Aedes vexans (Meigen) and Culex quinquefasciatus (Say). Sudan J Public Health. 2013; 8:164–70.
- Diallo D, Talla C, Ba Y, Dia I, Sall A, Diallo M. Temporal distribution and spatial pattern of abundance of the Rift Valley fever and West Nile fever vectors in Barkedji, Senegal. J Vector Ecol. 2011; 36(2):426– 36. https://doi.org/10.1111/j.1948-7134.2011.00184.x PMID: 22129415
- Huestis DL, Dao A, Diallo M, Sanogo ZL, Samake D, Yaro AS, et al. Windborne long-distance migration of malaria mosquitoes in the Sahel. Nature. 2019; 574(7778):404–8. Epub 2019/10/04. <u>https://doi.org/ 10.1038/s41586-019-1622-4</u> PMID: 31578527.
- Abdo-Salem S, Tran A, Grosbois V, Gerbier G, Al-Qadasi M, Saeed K, et al. Can Environmental and Socioeconomic Factors Explain the Recent Emergence of Rift Valley Fever in Yemen, 2000–2001? Vector Borne Zoonotic Dis. 2011; 11(6):773–9. <u>https://doi.org/10.1089/vbz.2010.0084</u> PMID: 21284504
- 69. Maquart M, Pascalis H, Abdouroihamane S, Roger M, Abdourahime F, Cardinale E, et al. Phylogeographic Reconstructions of a Rift Valley Fever Virus Strain Reveals Transboundary Animal Movements from Eastern Continental Africa to the Union of the Comoros. Transbound Emerg Dis 2016; 63(2): e281–5. https://doi.org/10.1111/tbed.12267 Epub 2014 Sep 12. PMID: 25213037
- **70.** Favier C, Chalvet-Monfray K, Sabatier P, Lancelot R, Fontenille D, Dubois M. Rift valley fever in West Africa: the role of space in endemicity. Trop Med Int Health. 2006; 11(12):1878–88. <u>https://doi.org/10.1111/j.1365-3156.2006.01746.x PMID: 17176353</u>